

University of Pennsylvania ScholarlyCommons

Neuroethics Publications

Center for Neuroscience & Society

12-2018

Biosocial Influences on Offending Across the Life Course

Olivia Choy University of Pennsylvania, oliviagchoy@gmail.com

Jill Portnoy University of Pennsylvania, jill.portnoy@gmail.com

Adrian Raine University of Pennsylvania, araine@sas.upenn.edu

Rheanna J. Remmel

Robert Schug

See next page for additional authors

Follow this and additional works at: https://repository.upenn.edu/neuroethics_pubs

Part of the Bioethics and Medical Ethics Commons, Neuroscience and Neurobiology Commons, and the Neurosciences Commons

Recommended Citation (OVERRIDE)

Choy, O., Portnoy, J., Raine, A., Remmel, R.J., Schug, R., Tuvblad, C., & Yang, Y. (2018). Biosocial Influences on Offending Across the Life Course. In D.P. Farrington, L. Kazemian, & A.R. Piquero (Eds.), The Oxford Handbook of Developmental and Life-Course Criminology. New York: Oxford University Press. [doi: 10.1093/oxfordhb/9780190201371.013.16]

This paper is posted at ScholarlyCommons. https://repository.upenn.edu/neuroethics_pubs/171 For more information, please contact repository@pobox.upenn.edu.

Abstract

This chapter presents major biological and biosocial findings in relation to the development of offending. It reviews empirical findings on the association between two psychophysiological factors, heart rate and skin conductance, and offending. The chapter then discusses the heritability of antisocial behavior and the contribution of genetics to the understanding of developmental trajectories, stability, and change in offending. The structural and functional brain abnormalities in antisocial individuals across different age groups are then discussed, along with research on hormones and neurotransmitters. Next, the chapter highlights the applications of neuropsychology in the understanding of offending. It concludes with potential areas for future research.

Keywords

biosocial influences, offending, heart rate, skin conductance, psychophysiological factors, antisocial behavior, genetics, brain abnormalities, neuropsychology

Disciplines

Bioethics and Medical Ethics | Neuroscience and Neurobiology | Neurosciences

Author(s)

Olivia Choy, Jill Portnoy, Adrian Raine, Rheanna J. Remmel, Robert Schug, Catherine Tuvblad, and Yaling Yang

Oxford Handbooks Online

Biosocial Influences on Offending Across the Life Course a

Olivia Choy, Jill Portnoy, Adrian Raine, Rheanna J. Remmel, Robert Schug, Catherine Tuvblad, and Yaling Yang

The Oxford Handbook of Developmental and Life-Course Criminology *Edited by David P. Farrington, Lila Kazemian, and Alex R. Piquero*

Print Publication Date: Jan 2019

Subject: Criminology and Criminal Justice, Life-Course and Developmental Criminology Online Publication Date: Dec 2018 DOI: 10.1093/oxfordhb/9780190201371.013.16

Abstract and Keywords

This chapter presents major biological and biosocial findings in relation to the development of offending. It reviews empirical findings on the association between two psychophysiological factors, heart rate and skin conductance, and offending. The chapter then discusses the heritability of antisocial behavior and the contribution of genetics to the understanding of developmental trajectories, stability, and change in offending. The structural and functional brain abnormalities in antisocial individuals across different age groups are then discussed, along with research on hormones and neurotransmitters. Next, the chapter highlights the applications of neuropsychology in the understanding of offending across the life span and reviews research on pre- and perinatal factors related to later offending. It concludes with potential areas for future research.

Keywords: biosocial influences, offending, heart rate, skin conductance, psychophysiological factors, antisocial behavior, genetics, brain abnormalities, neuropsychology

SINCE the late 20th century, the field of criminology has become increasingly aware of the contributions of biological sciences. Through longitudinal studies and research on different age groups, the role of biological factors in offending has been examined in a developmental context. Findings document that biological factors are associated with offending across the life span, although the strength of the associations may differ across development and between types of offenders. It is proposed that incorporating such factors into future developmental and life-course research and theories can lead to a better understanding of the etiology of offending.

In this chapter, major biological and biosocial findings in relation to the development of offending are presented. Offending is referred to as not only the violation of legal codes but also the broader spectrum of antisocial behavior. Section I reviews empirical findings on the association between two psychophysiological factors, heart rate and skin conduc-

Page 1 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

tance, and offending. Section II discusses the heritability of antisocial behavior and the contribution of genetics to the understanding of developmental trajectories, stability, and change in offending. Section III describes structural and functional brain abnormalities in antisocial individuals across different age groups. Section IV covers research on hormones and neurotransmitters. It examines the role of cortisol, testosterone, serotonin, and dopamine on offending. Section V highlights the applications of neuropsychology in the understanding of offending across the life span, particularly in the domains of verbal and spatial intelligence and executive functioning. Section VI reviews research on preand perinatal factors related to later offending, including prenatal alcohol, nicotine, and lead exposure, minor physical anomalies, and birth (p. 326) complications. Each of the six sections aims to address issues important to the developmental and life-course criminological literature including whether the biological factor is consistently associated with offending throughout the life course and whether persistent offenders differ from other offenders in terms of biological influences. Section VII concludes with potential areas for future research.

I. Psychophysiology

Psychophysiology is the study of cognition, behavior, and emotions as revealed through bodily events (Hugdahl 2001). Heart rate and skin conductance are psychophysiological measures that have been frequently examined in relation to offending.

A. Heart Rate

Heart rate is controlled by both the parasympathetic and sympathetic branches of the autonomic nervous system. A meta-analysis concluded that low resting heart rate is likely the best-replicated correlate of antisocial behavior in children and adolescents (Ortiz and Raine 2004). Although more commonly examined in youths, low resting heart rate is also a risk factor for antisocial behavior in adults (Lorber 2004; Armstrong et al. 2009; Portnoy and Farrington 2015), making low resting heart rate a biological risk factor for offending across the life course.

Importantly, low resting heart rate has been found to predict future levels of antisocial behavior in prospective longitudinal research (Farrington 1997; Raine, Venables, and Mednick 1997; Sijtsema et al. 2010; Jennings, Piquero, and Farrington 2013). One study found that low resting heart rate as young as at age 3 years predicted aggression at age 11 years (Raine et al., 1997). Findings from the Cambridge Study in Delinquent Development showed that low resting heart rate at age 18 years predicted offending up to age 50 years independent of covariates including smoking, sports participation, impulsivity, binge drinking, body mass index, and early childhood individual and environment risk factors (Jennings et al. 2013). This study demonstrated for the first time that the predictive utility of low resting heart rate could extend into late adulthood.

Page 2 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

It has also been found that low resting heart rate is only important in explaining initial levels rather than change in antisocial behavior over time. Baker et al. (2009) found that children with low resting heart rate at age 9 years were significantly more antisocial overall, but the reduction in antisocial behavior with age as the children entered early adolescence was not associated with heart rate. This suggests that low heart rate is a fixed, static neurobiological risk factor for antisocial behavior that does not predict desistance from offending throughout early adolescence.

Several theoretical explanations have been proposed to explain the relationship between resting heart rate and antisocial behavior. According to stimulation-seeking (p. 327) theory, low autonomic nervous system arousal is an unpleasant physiological state, leading those with low resting heart rates to seek stimulating behaviors, including antisocial behaviors, in order to increase their level of physiological arousal to a more optimal level (Quay 1965; Raine 2002a). An alternative theory argues that low resting heart rate may reflect a relative lack of fear, which could predispose some individuals to commit antisocial acts that require a degree of fearlessness to complete. Low heart rate could also impede early childhood fear conditioning to socializing punishments (Raine 1993, 2002a). While support for these two theories has been broadly found (Latvala et al. 2015), recent studies documented that stimulation seeking, but not fearlessness, mediated the relationship between low heart rate and antisocial behavior (Sijtsema et al. 2010; Portnoy et al. 2014). Thus, a stimulation-seeking mechanism may be more likely to underlie this relationship.

B. Skin Conductance

Skin conductance is an index of sympathetic nervous system activity that can be measured at rest or during laboratory tasks. Reduced skin conductance reactivity during fear conditioning paradigms has been associated with psychopathy (Birbaumer et al. 2005) and antisocial behavior, particularly persistent proactive aggression (Gao et al. 2015). Conditioning during childhood is thought to be central to socialization and conscience development. It has been suggested that the failure to condition could be a factor that predisposes some individuals to offend later in life (Eysenck 1977). Findings for skin conductance measured at rest tend to be less consistent. A meta-analysis found that low resting skin conductance was significantly associated with higher levels of psychopathy in adults and conduct problems in children, but not aggression or conduct problems in adolescence (Lorber 2004).

Like heart rate, reduced skin conductance has been documented to predict future levels of antisocial behavior. One study found that reduced skin conductance arousal at age 15 years predicted criminal behavior at age 24 years (Raine, Venables, and Williams 1990). Reduced skin conductance fear conditioning as early as age 3 years has been found to predict offending at age 23 years (Gao et al. 2010). These findings suggest that childhood and adolescent skin conductance can help to explain future levels of criminal behavior.

Page 3 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

C. Biosocial Interactions Involving Psychophysiology

Some studies have found that low resting heart rate combined with high social risk increases the likelihood of antisocial behavior (Raine et al. 2014). Similarly, skin conductance has been found to interact with social adversity to predict antisocial behavior, though patterns of interaction are not always consistent, with high skin conductance serving as a risk factor for antisocial behavior among children at high social risk in (p. 328) several studies (e.g., Cummings et al. 2007). In general, psychophysiological studies suggest a reduced pattern of autonomic arousal across the life course in antisocial individuals, although results may be partly dependent on the individual's social context.

II. Genetics

Compelling evidence from behavioral genetic research, which broadly includes twin, adoption, and family studies, shows that heritable influences also contribute to the development of offending. A review on 19 twin and adoption studies between ages 1 and 18 years found that heritability explained 65 percent and 48 percent of the variance in aggressive behavior and delinquent/rule-breaking behavior, respectively (Burt 2009). Additionally, summarizing results from 51 twin and adoption studies in children, adolescents, and adults, Rhee and Waldman (2002) found that genetic factors explained 41 percent of the variance in antisocial behavior. Given findings on the heritability of offending, molecular genetic research has identified candidate genes for offending. Lower monoamine oxidase A (MAOA) gene activity has been associated with violent behaviors and offending over the life course (e.g., Beaver et al. 2013). Other genes suggested to be associated with child and adult antisocial behavior include the catechol-O-methyltransferase (COMT) gene (Volavka, Bilder, and Nolan 2004; Hirata et al. 2013), the vasopressin receptor 1B (Zai et al. 2012) gene, the oxytocin receptor (OXTR) gene (Malik et al. 2012), the human dopamine transporter (DAT1) gene (Guo, Roettger, and Shih 2007), the D2 receptor polymorphism (DRD2) gene, and the D4 receptor polymorphism (DRD4) gene (Beaver et al. 2007; Boutwell et al. 2014).

Reviews have also found that aggressive and delinquent/rule-breaking behavior exhibit different etiological patterns across age (Burt 2009). Genetic influences on aggressive behavior increased across development, while shared environmental factors decreased. In contrast, delinquent/rule-breaking behavior showed a decrease in genetic influences across development, while shared environmental influences remained stable. The results show that aggressive behavior is primarily influenced by genetic factors, while delinquent/rule-breaking behavior is influenced by both genetic and shared environmental factors.

Since 2003, several studies have examined the genetic and environmental influences on psychopathic personality in children, adolescents, and adults (e.g., Viding et al. 2005; Viding, Frick, and Plomin 2007; Brook et al. 2010; Bezdjian et al. 2011; Hicks et al. 2012). According to the average twin correlations across these studies, the heritability of psy-

Page 4 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

chopathic personality in males is approximately 64 percent. For females, the heritability of psychopathic personality is approximately 48 percent (Tuvblad 2013). A longitudinal study reported that 58 percent and 62 percent of the stable variance in two features of psychopathic personality, fearless dominance and impulsive antisociality, respectively, from ages 17 to 24 years were explained by genetic factors (Blonigen et al. 2006; Tellegen and Waller 2008).

(p. 329) With regards to different developmental trajectories for offending, twin studies have shown that genetic influences are more important for stable-high/childhood-onset than for increasing/transitory antisocial behavior (e.g., Taylor, Iacono, and McGue 2000; Tuvblad et al. 2011). For example, in one study, genetic influences contributed more strongly to early-onset rather than late-onset delinquent behavior in 11-year-old boys (Taylor et al. 2000).

Longitudinal twin studies have also examined the contribution of genetics to stability in antisocial behavior. One study that measured antisocial behavior from ages 8 to 20 years showed that the stability of antisocial behavior was explained by a common latent antisocial behavior factor, for which genetics accounted for 67 percent of the variance (Tuvblad et al. 2011).

Other twin studies have examined the genetic contribution to change in antisocial behavior. This line of research has generally reported that change or "new" variance in antisocial behavior is primarily due to non-shared environmental factors (e.g., Haberstick et al. 2006). Analysis of three waves of data from the Minnesota Twin and Family Study showed that while genetic influences were to a large extent responsible for the initial level of antisocial personality disorder symptoms, non-shared environmental influences were largely responsible for change (Burt et al. 2007).

A. Biosocial Interactions Involving Genetic Factors

As in molecular genetic studies, such as that by Caspi et al. (2002), which found that childhood maltreatment led to violence in adulthood among individuals with low levels of MAOA expression, behavioral genetic studies have generated a large number of gene-environment interaction ($G \times E$) studies (for a review, see Dick 2011). Specifically, social factors such as family dysfunction, family warmth, high paternal punitive discipline, parental monitoring, religiosity, regional residency, and socioeconomic status have been found to moderate the genetic and environmental influences on antisocial behavior (Koopmans et al. 1999; Rowe, Almeida, and Jacobson 1999; Rose et al. 2001; Button et al. 2005; Tuvblad, Grann, and Lichtenstein 2006; Dick et al. 2007; Button et al. 2008; Middeldorp et al. 2014). Some studies have found higher heritability of antisocial behavior in individuals with low rather than high levels of social risk (Button et al. 2005), while others document that genetic influences contributed more to antisocial behavior when social risk was present (Dick et al. 2007).

Page 5 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

III. Brain Imaging

Regarding structural abnormalities associated with offending, studies have largely focused on regions involved in decision-making (e.g., prefrontal cortex), emotion regulation (e.g., amygdala, hippocampus), and reward-processing (e.g., striatum). One of (p. 330) the first structural brain imaging studies of antisocial adults documented an 11 percent reduction in gray matter volume in the prefrontal cortex of men with antisocial personality disorder compared to normal controls and a psychiatric control group (Raine et al. 2000). Yang et al. (2005, 2009, 2010a) found reduced gray matter volume and thickness in the middle frontal and orbitofrontal cortex and reduced volume and surface deformations in the amygdala in psychopaths with prior convictions (i.e., unsuccessful psychopaths) compared to psychopaths without convictions (i.e., successful psychopaths) and non-psychopathic controls. More recently, violent offenders were shown to have abnormal hippocampal structure compared to non-violent controls (Boccardi et al. 2010; Yang et al. 2010b). These frontal and limbic deficits were similarly found in a large sample of nearly 300 incarcerated criminal offenders (Ermer et al. 2012). Using vivo diffusion tensor magnetic resonance imaging tractography, Craig et al. (2009) further showed impaired amygdalaorbitofrontal connections in psychopaths with convictions. However, findings for the striatum are not conclusive, as some studies have documented enlarged putamen, nucleus accumbens, and caudate (Schiffer et al. 2011), while others showed smaller nucleus accumbens in offenders (Boccardi et al. 2013).

In addition to structural imaging research, functional imaging studies have presented evidence for impaired brain functioning in criminal offenders, especially in the prefrontal and temporal cortex. A meta-analysis by Yang and Raine (2009) of 43 studies revealed that increased antisocial behavior was associated with reduced prefrontal structure and function, particularly in the right orbitofrontal, left dorsolateral prefrontal, and right anterior cingulate cortex. Employing a neurocognitive task, the non-verbal Stroop task, Schiffer et al. (2014) found reduced function in the anterior cingulate, dorsolateral prefrontal, superior temporal, putamen, and amygdala in violent offenders with antisocial personality disorder compared to non-offenders. One recent study using resting-state functional magnetic resonance imaging (fMRI) also revealed reduced activity, measured by the amplitude of low-frequency fluctuation, in the right orbitofrontal cortex as well as the left temporal pole, right inferior temporal gyrus, and left cerebellum in these offenders with antisocial personality disorder (Liu et al. 2014). In line with these findings, Ly et al. (2012) found thinner cortices in the right inferior frontal cortex, anterior temporal cortex, and anterior cingulate cortex, which also corresponded to reduced functional connectivity between the left insula and left dorsal anterior cingulate cortex in psychopathic compared to non-psychopathic criminal offenders.

Recently, in a longitudinal study, males with lower amygdala volumes at age 26 years were found to exhibit increased aggression, violence, and psychopathic traits at a 3-year follow-up (Pardini et al. 2014). Similarly, in a study on adult male offenders, lower anteri-

Page 6 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

or cingulate activity was associated with a greater likelihood of rearrest (Aharoni et al. 2013). These findings suggest that brain deficits can predict later offending.

Consistent with research on adult offenders, studies of delinquent children and adolescents have revealed abnormal brain structures and function. For example, in a sample of female adolescents with conduct disorders, aggressive symptoms were negatively correlated with right dorsolateral prefrontal cortex volume, while (p. 331) callous-unemotional traits correlated positively with bilateral orbitofrontal cortex volume (Fairchild et al. 2013). Delinquents with high psychopathy scores also showed higher activity in the anterior cingulate cortex, insula, and amygdala during fear conditioning compared to healthy controls (Cohn et al. 2013). Longitudinally, a thicker temporal cortex was linked to higher rates of change in psychopathy during childhood and adolescence (Yang et al. 2015). Taken together, meta-analyses document that youths with disruptive behavior disorder or conduct problems show consistent functional deficits in the dorsal and rostral anterior cingulate and the medial prefrontal cortex (Alegria, Radua, and Rubia 2016), as well as reduced gray matter volume in the insula, left amygdala, and frontal and temporal regions (Rogers and De Brito 2016). One study documented that limbic structural deficits such as in the amygdala were associated with not only early-onset but also adolescent-onset conduct disorder (Fairchild et al. 2011). However, more recently, evidence has been found for quantitative differences in structural brain organization between childhood-onset and adolescence-onset conduct disorder (Fairchild et al. 2016) and for different growth trajectories of cortical thickness for distinct conduct problem pathways (Oostermeijer et al. 2016).

A. Biosocial Interactions Involving Brain Imaging

Although few brain imaging studies to date have addressed the role of psychosocial risk and protective factors on offending, several studies have begun to address two related issues concerning whether home background moderates the relationship between violence and brain functioning and whether brain deficits combine with psychosocial deficits to predispose one to violence. Regarding the first issue, two studies using brain functioning as the outcome variable have demonstrated a moderating effect of home background, but in opposing directions. In one study, murderers from non-deprived home backgrounds showed a 14.2 percent reduction in functioning of the right orbitofrontal cortex relative to murderers from deprived home backgrounds characterized by abuse, neglect, and marital violence (Raine et al. 1998). It was argued that neurobiological deficits are more pronounced among violent individuals who lack the psychosocial deprivation that normally provides a "social push" toward violence. In contrast, a second fMRI study showed that violent offenders who had been severely abused as children were more likely to show poor temporal lobe functioning compared to violent offenders lacking abuse (Raine et al. 2001).

Turning to the second issue, using violence as an outcome variable, an anatomical magnetic resonance imaging study of individuals with antisocial personality disorder and high psychopathy scores showed that the combination of reduced prefrontal gray volume, low

Page 7 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

autonomic responsivity, and a set of 10 psychosocial deficits correctly classified 88.5 percent of subjects into antisocial personality disorder or control groups (compared to 73 percent for psychosocial predictors only and 76.9 percent for biological predictors only; Raine et al. 2000). A second structural imaging study on the corpus callosum in psychopaths showed that the combination of psychosocial risk factors (p. 332) with callosal measures accounted for 81.5 percent of the variance (Raine et al. 2003). Structural brain measures accounted for a significant increase in the variance in psychopathic/antisocial behavior over and above psychosocial risk factors in both studies.

IV. Hormones and Neurotransmitters

Compared to brain imaging research, fewer studies have examined the relationship between hormones and offending. Two most frequently studied hormones in relation to antisocial behavior are cortisol and testosterone, regulated by the hypothalamus-pituitaryadrenal (HPA) axis and hypothalamus-pituitary-gonadal (HPG) axis, respectively.

A. Cortisol

Studies in children and adolescents have shown that cortisol may be related to antisocial behavior early in life. One meta-analysis found a mean effect size of d = -.40 for the relationship between basal cortisol levels and disruptive behavior or aggressive symptoms in children after study sample sizes were taken into account (van Goozen et al. 2007). The mean effect size for cortisol reactivity in response to a stressor across 4 studies after correcting for sample sizes was d = .42. A second meta-analysis on 72 study outcomes found that in preschoolers (aged 0 to 5 years), higher basal cortisol was associated with externalizing behavior (d = .18). Low basal cortisol was associated with externalizing behavior (d = .28) in elementary school-aged children (aged 5 to 12 years; Alink et al. 2008). However, no significant association was found between basal cortisol and externalizing behavior in adolescents or between cortisol reactivity to stress and externalizing behavior.

Similar findings were obtained for adults as low cortisol levels were found in offenders with psychopathy compared to non-psychopathic offenders (Holi et al. 2006; Cima, Smeets, and Jelicic 2008). In a study on cortisol reactivity using a social stressor, a significant difference in cortisol levels from pre- to post-stressor was observed, but only in males with low rather than high levels of psychopathy (O'Leary, Loney, and Eckel 2007).

Several studies have examined cortisol levels in relation to different categories of offenders. Fairchild et al. (2008) found that basal cortisol level or cortisol reactivity to a stressor did not differ between male adolescents with early-onset and adolescence-onset conduct disorder. It has been proposed that structural abnormalities in the amygdala may underlie the finding that early- and adolescent-onset conduct disorder are associated with lower cortisol responses to stress as the amygdala is involved in initiating HPA responses to stress (Fairchild et al. 2011). However, another (p. 333) study on boys aged 7 to 12 years

Page 8 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

found that lower cortisol was more strongly related to persistently aggressive boys as well as those with childhood-onset conduct disorder compared to adolescence-onset conduct disorder (McBurnett et al. 2000). This is bolstered by findings that low basal cortisol levels predicted disruptive behavior in boys and girls only if conduct problems were already present at age 10 to 12 years (Sondeijker et al. 2008) and that persistently high-aggressive adolescents exhibited decreased cortisol levels consistently over time compared to low-aggressive adolescents (Platje et al. 2013). Such findings suggest that cortisol levels are related to the persistent trajectory of antisocial behavior rather than the prediction of the onset of behavior problems at later ages.

B. Testosterone

Experimental studies have shown that increased testosterone levels are associated with increased levels of aggression (Pope, Kouri, and Hudson 2000) and decreased levels of empathy (van Honk et al. 2011), which are associated with offending. Associations between higher levels of testosterone and antisocial behavior have been reported in children and adolescents. For example, adolescents with high levels of externalizing behaviors have been documented to have higher levels of testosterone than individuals with low levels of externalizing behaviors (Maras et al. 2003). Furthermore, testosterone levels were found to be higher in a disruptive behavior disorder group than in normal controls for an older subset of participants (aged 9 to 11 years) compared to those of younger ages (aged 5 to 8 years; Chance et al. 2000). However, some other studies have found mixed results on the testosterone-antisocial behavior relationship among children and adolescents (e.g., van Goozen et al. 1998; Dorn et al. 2009).

Additional studies have found a positive relationship between testosterone levels in adulthood and retrospectively reported severity of conduct disorder symptoms in childhood (e.g., Mazur 1995). Longitudinal studies have documented that testosterone levels in a community sample at ages 12 and 14 years predicted antisocial norm-violating behaviors at age 16 years (Tarter et al. 2009), and testosterone levels in conduct disordered boys at age 13 years predicted delinquency and criminal behavior at ages 16 and 21 years (Van Bokhoven et al. 2006). In adults, meta-analytic evidence suggests that testosterone is positively associated with aggression, with higher associations found for the age group of 22 to 35 years and in offender compared to non-offender populations (Archer et al. 2005). Higher testosterone levels also correlated positively with psychopathy scores in convicted criminals (Stålenheim et al. 1998). On the other hand, Glenn et al. (2011) found that instead of baseline testosterone, the ratio of baseline testosterone to cortisol reactivity after stress was significantly related to psychopathy. Generally, research suggests there is a small positive correlation between testosterone and antisocial behavior throughout the life span. Relationships seem to be weakest in young children and get stronger as individuals age (Yildirim and Derksen 2012).

Page 9 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

(p. 334) C. Serotonin and Dopamine

Serotonin and dopamine are neurotransmitters that have been implicated in antisocial behavior and specifically in psychopathy. Most commonly, researchers have examined neurotransmitter metabolite levels in cerebrospinal fluid, such as HVA, a metabolite of dopamine, and 5-HIAA, a metabolite of serotonin (Freedman and Verdun-Jones 2010).

Studies have found that examining serotonin and dopamine levels together provides a better prediction of psychopathy scores. In a sample of violent offenders, the ratio of HVA to 5-HIAA was positively associated with psychopathy scores, particularly the Factor 2 Antisocial/Lifestyle score, which has been linked to life-course-persistent offending (Soderstrom et al. 2001; Yildirim and Derksen 2012). In a follow-up study, these results were replicated in a forensic sample. The HVA:5-HIAA ratio was positively related to childhood-onset disruptive disorders (Soderstrom et al. 2003).

Other research has suggested that serotonin levels are generally low in antisocial populations. One meta-analysis on 20 reports revealed reduced 5-HIAA in antisocial populations, particularly for individuals younger than 30 years, supporting the possibility that age-related increases in serotonin correlate with age-related declines in crime (Moore, Scarpa, and Raine 2002). Significantly lower serotonin levels were also found in boys with high levels of callous-unemotional traits (Moul et al. 2013). In children and adolescents with obsessive-compulsive disorder, participants with comorbid disruptive behavior disorders had significantly lower blood serotonin concentrations than participants with no comorbid behavior disorder (Hanna, Yuwiler, and Coates 1995). In the same study, a negative relationship was also found between serotonin concentration and externalizing and aggression scores. Additionally, reduced concentrations of somatostatin, a peptide which stimulates the release of serotonin, have been found in the cerebrospinal fluid of children with disruptive behavior disorders compared to children with obsessive-compulsive disorder (Kruesi et al. 1990). Although reduced serotonin levels have been found in children, findings suggest that the strongest serotonin effects on offending occur in young adulthood (Moore et al. 2002). All in all, despite some mixed findings (Hughes et al. 1996), there is reasonably strong evidence that serotonin and dopamine play a role in the development of offending.

D. Biosocial Interactions Involving Hormones and Neurotransmitters

Limited research has been conducted on biosocial interactions involving hormones and neurotransmitters. In one study, maltreatment was a significant moderator of the cortisol dysregulation-antisocial behavior relationship, such that low cortisol levels were more strongly associated with antisocial behavior in nonmaltreated children compared to maltreated peers (Hawes, Brennan, and Dadds 2009). In a similar vein, Hawes et al. (p. 335) (2009) suggested that early adversity plays a role in the development of antisocial behavior in children with low levels of callous-unemotional traits and higher basal cortisol levels, while high levels of callous-unemotional traits and low basal cortisol levels characterize a particularly severe subgroup for whom antisocial behavior develops somewhat inde-

Page 10 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

pendently of social adversity. A study on Dutch adolescents documented that among individuals who experienced low levels of an environmental stressor, namely neighborhood density, lower cortisol activity significantly predicted higher levels of delinquency and aggression (Yu et al. 2016). Thus, HPA-axis dysfunction may play a more significant role in the development of chronic antisocial behavior for individuals who have not been exposed to adversity.

V. Neuropsychological Factors

Neuropsychology, the indirect, behavior-based assessment of brain dysfunction, has also been used to understand offending across the life span. Neuropsychological investigations of various forms of antisociality have largely targeted deficits in specific domains of cognitive functioning such as verbal and spatial intelligence and executive abilities.

A. Verbal and Spatial Intelligence

To date, the best-replicated cognitive correlate of antisocial, violent, and criminal behavior among non-mentally ill individuals is general intelligence (e.g., IQ or Full Scale IQ) deficits (Wilson and Herrnstein 1985). Reduced verbal relative to spatial/performance IQ —a possible marker for left hemispheric dysfunction—has generally been documented to characterize both males and females from different age groups across studies of antisocial individuals (Raine 1993; Isen 2010). However, some antisocial individuals, such as those with antisocial personality disorder and psychopathy, have not consistently shown intellectual performance or verbal intelligence deficits (Barkataki et al. 2006; Kosson et al. 2007), although relationships have been noted between some specific psychopathic traits (i.e., criminal versatility and violence) and verbal dysfunction (Rasmussen, Almvik, and Levander 2001). Thus, while global and/or verbal intellectual dysfunction may characterize adult antisocial individuals in general, they may not characterize specific constellations of criminogenic and antisocial traits.

Reduced verbal intelligence also appears largely characteristic of antisocial children and adolescents (e.g., Barker et al. 2007). Moffitt, Lynam, and Silva (1994) found that verbal deficits in early adolescence predicted delinquency in later adolescence for persistent, high-level offenders who began offending in pre-adolescence. However, mixed results have been found for juvenile psychopathy. Loney et al. (1998) found no verbal deficits in children with conduct problems and callous-unemotional traits, (p. 336) while Salekin et al. (2004) found that verbal intelligence was positively related to the superficial and deceitful interpersonal style traits and inversely related to the affective processing-disturbance traits of psychopathy in juvenile inmates. In summary, verbal deficits in populations of antisocial youth overall appear relatively consistent, though continued studies of psychopathic youth may assist in clarifying heterogeneous verbal IQ findings among antisocial juveniles as in adults.

Page 11 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Longitudinal studies of community-based samples may call into question the classic view of verbal but not performance IQ deficits in antisocial individuals. In a Pittsburgh youth sample including childhood-limited, adolescent-limited, and life-course-persistent offenders, Raine et al. (2005) found both spatial and verbal impairments. In another sample from Mauritius, Raine et al. (2002c) found early spatial but not verbal deficits at age 3, and later spatial and verbal deficits at age 11 in persistently antisocial individuals. These results suggest that while early spatial deficits contribute to persistent antisocial behavior, verbal deficits may be developmentally acquired. Results support a proposed early starter spatial impairment model of life-course offending, in which early deficits in visuospatial functioning may interfere with mother-infant bonding, possibly reflecting right hemispheric dysfunction that disrupts emotional processing and regulation and, in turn, contributes to persistent offending.

B. Executive Functioning

Executive functioning deficits are thought to represent impairment in frontal lobe functioning and are indicated by performance errors on neuropsychological tests of strategy formation, cognitive flexibility, or impulsivity (i.e., category, maze-tracing, Stroop interference, card sorting, verbal fluency and tower tests, and go/no-go and gambling tasks). In Morgan and Lilienfeld's (2000) classic quantitative review of 39 studies, overall executive functioning deficits were observed in antisocial individuals compared to controls. Strongest effects were found for the Porteus Mazes test and antisociality defined by judicial status. More recently, executive dysfunction has been associated with aggressive, violent, and antisocial personality-disordered populations (e.g., Stanford et al. 2007; Hancock, Tapscott, and Hoaken 2010; Dolan 2012), property criminality (Barker et al. 2007), child molesters with and without pedophilia (Schiffer and Vonlaufen 2011), single as opposed to multiple homicide victims in indigent murder defendants and death row inmates (Hanlon et al. 2010), murderers with schizophrenia compared to non-violent men with schizophrenia (Hanlon et al. 2012), mentally challenged versus non-impaired forensic hospital patients (Bastert et al. 2012), and offenders characterized by reactive as opposed to instrumental violence (Broomhall 2005).

Psychopathy in adults has not been consistently associated with general executive functioning deficits (e.g., Kosson et al. 2007). Some neuropsychological studies have shown that psychopathy may be characterized more by orbitofrontal dysfunction, which is associated with processing rewards and punishments, and emotion (p. 337) (Rolls 2000; Blair et al. 2006). Additionally, successful, uncaught psychopaths have demonstrated significantly better dorsolateral prefrontal task performance relative to unsuccessful psychopaths and controls (Ishikawa et al. 2001), while white-collar criminals have been found to show increased executive functioning compared to offender controls (Raine et al. 2012). Furthermore, violent antisocial personality disordered offenders with and without psychopathy have demonstrated similar deficits in terms of "cool executive functioning," namely top-down processes subsumed by the dorsolateral and ventrolateral prefrontal cortex, that are distinctly cognitive in nature, such as working memory, response inhibition, planning, sustained attention, and attentional set-shifting, and "hot executive func-

Page 12 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

tioning," namely processes with an affective, motivational, or incentive/reward component subsumed by ventromedial connections between the mesolimbic reward pathway and the ventromedial prefrontal cortex, such as appraisal of the motivational significance of a stimulus in emotional decision-making (De Brito et al. 2013).

Findings on children and adolescents have been more mixed, with executive functioning deficits characterizing some antisocial youths (e.g., Cauffman et al. 2005) and not others (Moffitt et al. 1994; Nigg et al. 2004). The development of executive functions along with the ongoing myelination of the frontal cortex into and beyond adolescence (Raine 2002b) may explain differential patterns of executive functioning deficits among children and adults. This is supported by findings of executive functioning impairments in older maximum security hospital patients (Nestor 1992) and more pronounced impairments on an orbitofrontal neuropsychological task in psychopathic adults than psychopathic children (Blair 2006).

C. Biosocial Interactions Involving Neuropsychology

In a study examining biosocial interactions, Gao et al. (2009) found that neurocognitive deficits indicated by more risky decision-making in the Iowa Gambling Task were associated with psychopathic tendencies only in children with higher socioeconomic status. Additionally, progressive cognitive dysfunction affected by adverse psychosocial experience may explain early-onset antisocial behavior (Aguilar et al. 2000) and lifetime, cumulative biosocial risk interactions may be stronger predictors of persistent aggression than risks only occurring in childhood or adolescence (Brennan et al. 2003). More specifically, Brennan et al. (2003), in a study of 370 Australian adolescents, identified that an interaction of biological risk factors including neuropsychological deficits and social risk factors predicted life-course-persistent aggression in boys and girls. Alternatively, the late-developing prefrontal cortex may be overloaded by the social and executive functioning demands of late adolescence, possibly leading to prefrontal dysfunction, behavioral inhibition failure, and significantly increased antisocial behavior (Raine 2002b). In sum, the neuropsychological literature demonstrates how the study of behavioral expressions of brain dysfunction has informed developmental neurobiological perspectives of offending across the life span.

(p. 338) VI. Early Health Risks

Evidence suggests that risk factors experienced early in life, such as during the prenatal and perinatal periods of development, are associated with longitudinal patterns of offending and may lead to the most detrimental effects over the life span (Day, Wanklyn, and Yessine 2014). Prenatal and perinatal factors that have been most closely linked to antisocial behavior include prenatal nicotine, alcohol, and lead exposure, minor physical anomalies, and birth complications.

Page 13 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

A. Prenatal Nicotine, Alcohol, and Lead Exposure

Children who are exposed to maternal smoking during pregnancy have been documented to have an elevated risk of offending throughout the life course (Wakschlag et al. 2002). Numerous studies have found associations between prenatal maternal smoking and juvenile offending, delinquency, conduct disorder, and violent offending (e.g., Wakschlag et al. 1997; Brennan, Grekin, and Mednick 1999; Paradis et al. 2015). In particular, a dose-response relationship was observed between the degree of prenatal maternal smoking and the extent of offspring's nonviolent and violent offending assessed at age 34 years (Brennan et al. 1999). However, there is current debate regarding whether the nicotine exposure-offending association involves a genetic confound (Glenn and Raine 2014).

Prenatal exposure to alcohol results in cognitive, behavioral, social, and physical deficits and can lead to a diagnosis of fetal alcohol syndrome (FAS). Fetal alcohol exposure has been documented as a risk factor for antisocial behavior in children, adolescents, and adults (Olson et al. 1997; Fast, Conry, and Loock 1999). However, even without FAS, high rates of delinquency have been found in children and adolescents with heavy fetal alcohol exposure (Mattson and Riley 2000).

Besides nicotine and alcohol exposure, a prospective study found that prenatal maternal blood lead concentrations during the first and second trimesters of pregnancy were associated with higher rates of criminal arrests measured at ages 19 to 24 years (Wright et al. 2008). Another longitudinal study on 195 adolescents found that lead levels from the prenatal period to 6.5 years of age were associated with delinquent and antisocial behavior in middle adolescence (Dietrich et al. 2001). Although few longitudinal studies in this area exist, these studies demonstrate that prenatal lead exposure is associated with the development of offending.

B. Minor Physical Anomalies

Minor physical anomalies (MPAs) such as low-seated ears, single palmar crease, and furrowed tongue are considered indicators of fetal neural maldevelopment near the end (p. 339) of the first trimester or the beginning of the second trimester of pregnancy (Firestone and Peters 1983). Studies have found that MPAs are associated with greater antisocial behavior in children, adolescents, and adults, particularly for violent as opposed to non-violent offending (Glenn and Raine 2014). For example, MPAs measured at age 14 years predicted violent delinquency in 170 males at age 17 years, independent of childhood physical aggression or family adversity (Arseneault et al. 2000). Other studies have found that a larger number of MPAs is associated with recidivistic violent criminal behavior. One study documented that recidivistic violent offenders at ages 20 to 22 years had more MPAs between ages 11 to 13 years compared to individuals with one or fewer violent offenses (Kandel et al. 1989).

Page 14 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

C. Birth Complications

Birth complications, such as pre-eclampsia, preterm birth, and breech fetal positioning, have also been found to predispose to later offending (e.g., Liu et al. 2009). For example, Kandel and Mednick (1991) found that high delivery complications were associated with adult violent offending. Additionally, findings from a longitudinal study, the Fragile Families and Child Well-Being Study, showed that low birthweight was linked to serious aggression and destructive behavior at age 5 years and the relationship was mediated by verbal skills (Vaske, Newsome, and Boisvert 2013). Other perinatal risk factors for later offending include being small for gestational age and a small head circumference (Babchishin et al. 2017).

D. Biosocial Interactions Involving Early Health Risks

Studies have documented that prenatal nicotine exposure, MPAs, and birth complications interact with social factors to predispose to later offending. For example, prenatal nicotine exposure was found to lead to an 11.9-fold and 14.2-fold increase in recidivistic violent offending in adulthood when combined with the individual social risk factor of being raised in a single-parent family and with a group of psychosocial risk factors, respectively (Räsänen et al. 1999). Increased risk has particularly been observed for persistent violent offending (Brennan et al. 1999; Gibson and Tibbetts 2000; Brennan et al. 2002). Moreover, MPAs in boys at age 12 years were related to violent, but not non-violent property offending at age 21 years, but only among individuals reared in unstable homes (Mednick and Kandel 1988). Similarly, Brennan, Mednick, and Raine (1997) and Pine et al. (1997) found higher rates of adult violent crime in males and greater risk for disruptive behavior and conduct disorder at age 17 years among individuals with both MPAs and social risk factors. A recent study documented that individuals born at low birthweight were at an increased risk of adult offending if they were born to adolescent mothers (Vaske et al. 2015). This is consistent with the finding that birth complications combined with early maternal rejection measured at (p. 340) age 1 year increased the likelihood of violent offending at ages 18 and 34 years (Raine, Brennan, and Mednick 1994, 1997). Birth complications have also been found to interact with other psychosocial factors such as poor parenting (Hodgins, Kratzer, and McNeil 2001), family adversity (Arseneault et al. 2002), and being an only child (Kemppainen et al. 2001) to lead to adult violent offending. These studies suggest that increased offending is observed when both early health risks and environmental risk factors are present.

VII. Conclusion

Through a review of extant research, this chapter sheds light on the development of antisocial behavior and risk factors for offending at different ages, issues central to developmental and life-course criminology. Despite some null findings, many biological risk factors such as autonomic underarousal, genetics, structural and functional brain abnormalities (particularly in the prefrontal and temporal cortex), low basal cortisol, high testos-

Page 15 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

terone, low serotonin, neuropsychological deficits, and early health risks are associated with antisocial behavior in children, adolescents, and adults. The strength of the risk factor-antisocial behavior associations may differ across development. For example, genetic influences increased across development for aggressive behavior and decreased across development for delinquent/rule-breaking behavior. Brain deficits predisposed individuals to more severe antisocial behavior if damage occurred earlier rather than later in life. In terms of hormones and neurotransmitters, cortisol levels were found to decrease across development for aggressive individuals. The relationship between high testosterone and antisocial behavior was weakest in young children and strongest in adults, while the effect of low serotonin on antisocial behavior was strongest in young adulthood. Furthermore, antisocial adults seem to suffer a greater degree of executive functioning deficits than younger antisocial populations. Studies examining biosocial interactions have also found that antisocial individuals exposed to fewer social stressors are more likely to exhibit biological risk factors compared to those with high social risk and that individuals are most likely to offend over the life course when both social and biological risk factors are present. Such interaction effects have been found in relation to child, adolescent, and adult offending.

The studies also revealed some differences in the associations between biological factors and offending for different types of offenders. For example, genetic influences were more important for childhood-onset compared to late-onset offending, brain deficits were associated with both early- and adolescent-onset offending, and a higher ratio of HVA to 5-HIAA was associated with life-course-persistent and child-onset offending. Additionally, it has been suggested that low heart rate, early spatial deficits, and perinatal complications may contribute particularly to life-course-persistent offending.

(p. 341) A. Future Directions

Despite these findings, greater research is needed to advance understanding of the role of biology in developmental and life-course criminology. One area of future research involves protective factors for offending. Although psychological and social factors such as attachment (Farrington 2005a) and life events (Wikström 2005) have been proposed in developmental and life-course theories as variables that inhibit offending, such theories do not consider the role of biological factors as possible protective factors. Studies such as that by Raine, Venables, and Williams (1995, 1996) have documented that high resting heart rate and skin conductance can serve as protective factors. For example, antisocial adolescents who desisted from adult crime had significantly better skin conductance conditioning at age 15 years than persisters who were criminal at age 29 years (Raine, Venables, and Williams 1996). Nonetheless, research on biological protective factors is very much more limited compared to that on risk factors. Additional research can provide much-needed insight on the topic of desistance.

Developmental and life-course criminology is also concerned with the effects of life events on the development of offending (Farrington 2005b). Recent findings have provided support for a social neurocriminology perspective, in which biological factors are influ-

Page 16 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

enced by social environmental processes to affect antisocial behavior, by documenting that low heart rate partly mediated the social adversity-antisocial behavior relationship (Choy et al. 2015; Fagan, Zhang, and Gao 2017). Given that marriage is associated with lower levels of testosterone in males (Gray et al. 2002) and the finding that adolescents who experience adversity in the form of maltreatment early in life exhibited lower gray matter volumes in corticostriatal-limbic regions such as the dorsolateral prefrontal cortex and amygdala (Edmiston et al. 2011), future efforts could be directed at understanding how social environmental factors can affect offending through changes in biology.

In addition, although several developmental models such as that of Moffitt (1993) and Lahey and Waldman (2005) recognize some biological influences, they do not necessarily emphasize the interactive effects between biological and psychosocial variables. Many theories still fail to incorporate biological factors such as genetics or hormones in understanding the etiology of offending (Barnes et al. 2014). More biological testing should be conducted particularly in prospective longitudinal studies to examine within-individual differences in offending and to investigate biosocial interaction effects. In light of the proposed notion that criminological variables are affected by genetic influences, longitudinal research can be beneficial in accounting for some genetic influences as respondents serve as their own control over many observation points (Barnes et al. 2014). Such efforts can greatly contribute to a better understanding of the role of biology in models of offending and pave the way for the development of early intervention and prevention strategies for crime reduction.

References

Aguilar, Benjamin, L. Alan Sroufe, Byron Egeland, and Elizabeth Carlson. 2000. "Distinguishing the Early-Onset/Persistent and Adolescence-Onset Antisocial Behavior Types: From Birth to 16 Years." *Development and Psychopathology* 12: 109–132.

Aharoni, Eyal, Gina M. Vincent, Carla L. Harenski, Vince D. Calhoun, Walter Sinnott-Armstrong, Michael S. Gazzaniga, and Kent A. Kiehl. 2013. "Neuroprediction of Future Rearrest." *Proceedings of the National Academy of Sciences* 110: 6223–6228.

Alegria, Analucia A., Joaquim Radua, and Katya Rubia. 2016. "Meta-Analysis of fMRI Studies of Disruptive Behavior Disorders." *The American Journal of Psychiatry* 173: 1119– 1130.

Alink, Lenneke R. A., Marinus H. van IJzendoorn, Marian J. Bakermans-Kranenburg, Judi Mesman, Femmie Juffer, and Hans M. Koot. 2008. "Cortisol and Externalizing Behavior in Children and Adolescents: Mixed Meta-Analytic Evidence for the Inverse Relation of Basal Cortisol and Cortisol Reactivity with Externalizing Behavior." *Developmental Psychobiology* 50: 427–450.

Archer, John, Nicola Graham-Kevan, and Michelle Davies. 2005. "Testosterone and Aggression: A Reanalysis of Book, Starzyk, and Quinsey's (2001) Study." *Aggression and Violent Behavior* 10: 241–261.

Page 17 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Armstrong, Todd A., Shawn Keller, Travis W. Franklin, and Scott N. MacMillan. 2009. "Low Resting Heart Rate and Antisocial Behavior: A Brief Review of Evidence and Preliminary Results from a New Test." *Criminal Justice and Behavior* 36: 1125–1140.

Arseneault, Louise, Richard E. Tremblay, Bernard Boulerice, and Jean-François Saucier. 2002. "Obstetrical Complications and Violent Delinquency: Testing Two Developmental Pathways." *Child Development* 73: 496–508.

Arseneault, Louise, Richard E. Tremblay, Bernard Boulerice, Jean R. Séguin, and Jean-François Saucier. 2000. "Minor Physical Anomalies and Family Adversity as Risk Factors for Violent Delinquency in Adolescence." *American Journal of Psychiatry* 157: 917–923.

Babchishin, Kelly M., Michael C. Seto, Amir Sariaslan, Paul Lichtenstein, Seena Fazel, and Niklas Långström. 2017. "Parental and Perinatal Risk Factors for Sexual Offending in Men: A Nationwide Case-Control Study." *Psychological Medicine* 46: 305–315.

Barkataki, Ian, Veena Kumari, Mrigendra Das, Pamela Taylor, and Tonmoy Sharma. 2006. "Volumetric Structural Brain Abnormalities in Men with Schizophrenia or Antisocial Personality Disorder." *Behavioural Brain Research* 169: 239–247.

Baker, Laura A., Catherine Tuvblad, Chandra Reynolds, Mo Zheng, Dora Isabel Lozano, and Adrian Raine. 2009. "Resting Heart Rate and the Development of Antisocial Behavior from Age 9 to 14: Genetic and Environmental Influences." *Development and Psychopathology* 21: 939–960.

Barker, Edward D., Jean R. Séguin, Helene Raskin White, Marsha E. Bates, Eric Lacourse, Rene Carbonneau, and Richard E. Tremblay. 2007. "Developmental Trajectories of Male Physical Violence and Theft: Relations to Neurocognitive Performance." *Archives of General Psychiatry* 64: 592–599.

Barnes, J. C., Brian B. Boutwell, Kevin M. Beaver, Chris L. Gibson, and John P. Wright. 2014. "On the Consequences of Ignoring Genetic Influences in Criminological Research." *Journal of Criminal Justice* 42: 471–482.

Bastert, Eva, Detlef Schläfke, Annika Pein, Franziska Kupke, and Jörg M. Fegert. 2012. "Mentally Challenged Patients in a Forensic Hospital: A Feasibility Study Concerning the Executive Functions of Forensic Patients with Organic Brain Disorder, Learning Disability, or Mental Retardation." *International Journal of Law and Psychiatry* 35: 207–212.

(p. 343) Beaver, Kevin M., J. C. Barnes, and Brian B. Boutwell. 2013. "The 2-Repeat Allele of the MAOA Gene Confers an Increased Risk for Shooting and Stabbing Behaviors." *Psychiatric Quarterly*, 85: 257–265.

Beaver, Kevin M., John Paul Wright, Brian B. Boutwell, J. C. Barnes, Matt DeLisi, and Michael G. Vaughn. 2013. "Exploring the Association between the 2-Repeat Allele of the MAOA Gene Promoter Polymorphism and Psychopathic Personality Traits, Arrests, Incar-

Page 18 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

ceration, and Lifetime Antisocial Behavior." *Personality and Individual Differences* 54: 164–168.

Beaver, Kevin M., John Paul Wright, Matt DeLisi, Anthony Walsh, Michael G. Vaughn, Danielle Boisvert, and Jamie Vaske. 2007. "A Gene x Gene Interaction Between Drd2 and Drd4 Is Associated with Conduct Disorder and Antisocial Behavior in Males." *Behavioral and Brain Functions* 3: 1186–1193.

Bezdjian, Serena, Catherine Tuvblad, Adrian Raine, and Laura A. Baker. 2011. "The Genetic and Environmental Covariation Among Psychopathic Personality Traits, and Reactive and Proactive Aggression in Childhood." *Child Development* 82: 1267–1281.

Birbaumer, Niels, Ralf Veit, Martin Lotze, Michael Erb, Christiane Hermann, Wolfgang Grodd, and Herta Flor. 2005. "Deficient Fear Conditioning in Psychopathy: A Functional Magnetic Resonance Imaging Study." *Archives of General Psychiatry* 62: 799–805.

Blair, R. J. R. 2006. "The Emergence of Psychopathy: Implications for the Neuropsychological Approach to Developmental Disorders." *Cognition* 101: 414–442.

Blair, K. S., Chris Newman, Derek GV Mitchell, R. A. Richell, Alan Leonard, John Morton, and R. J. R. Blair. 2006. "Differentiating Among Prefrontal Substrates in Psychopathy: Neuropsychological Test Findings." *Neuropsychology* 20: 153–165.

Blonigen, Daniel M., Brian M. Hicks, Robert F. Krueger, Christopher J. Patrick, and William G. Iacono. 2006. "Continuity and Change in Psychopathic Traits as Measured via Normal-Range Personality: A Longitudinal-Biometric Study." *Journal of Abnormal Psychology* 115: 85–95.

Boccardi, Marina, Martina Bocchetta, Hannu J. Aronen, Eila Repo-Tiihonen, Olli Vaurio, Paul M. Thompson, Jari Tiihonen, and Giovanni B. Frisoni. 2013. "Atypical Nucleus Accumbens Morphology in Psychopathy: Another Limbic Piece in the Puzzle." *International Journal of Law and Psychiatry* 36: 157–167.

Boccardi, Marina, Rossana Ganzola, Roberta Rossi, Francesca Sabattoli, Mikko P. Laakso, Eila Repo-Tiihonen, Olli Vaurio, et al. 2010. "Abnormal Hippocampal Shape in Offenders with Psychopathy." *Human Brain Mapping* 31: 438–447.

Boutwell, Brian B., Scott Menard, J. C. Barnes, Kevin M. Beaver, Todd A. Armstrong, and Danielle Boisvert. 2014. "The Role of Gene-Gene Interaction in the Prediction of Criminal Behavior." *Comprehensive Psychiatry* 55: 483–488.

Brennan, Patricia A., Emily R. Grekin, and Sarnoff A. Mednick. 1999. "Maternal Smoking During Pregnancy and Adult Male Criminal Outcomes." *Archives of General Psychiatry* 56: 215–219.

Brennan, Patricia A., Emily R. Grekin, Erik Lykke Mortensen, and Sarnoff A. Mednick. 2002. "Relationship of Maternal Smoking During Pregnancy with Criminal Arrest and

Page 19 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Hospitalization for Substance Abuse in Male and Female Adult Offspring." *American Journal of Psychiatry* 159: 48–54.

Brennan, Patricia A., Jason Hall, William Bor, Jake M. Najman, and Gail Williams. 2003. "Integrating Biological and Social Processes in Relation to Early-Onset Persistent Aggression in Boys and Girls." *Developmental Psychology* 39: 309–323.

Brennan, Patricia A., Sarnoff A. Mednick, and Adrian Raine. 1997. "Biosocial Interactions and Violence: A Focus on Perinatal Factors." In *Biosocial Bases of Violence*, edited by (p. 344) Adrian Raine, Patricia A. Brennan, David P. Farrington, and Sarnoff A. Mednick, 163–174. New York: Plenum.

Brook, Michael, Matthew S. Panizzon, David S. Kosson, Elizabeth A. Sullivan, Michael J. Lyons, Carol E. Franz, Seth A. Eisen, and William S. Kremen. 2010. "Psychopathic Personality Traits in Middle-Aged Male Twins: A Behavior Genetic Investigation." *Journal of Personality Disorders* 24: 473–486.

Broomhall, Luke. 2005. "Acquired Sociopathy: A Neuropsychological Study of Executive Dysfunction in Violent Offenders." *Psychiatry, Psychology and Law* 12: 367–387.

Burt, S. Alexandra. 2009. "Are There Meaningful Etiological Differences within Antisocial Behavior? Results of a Meta-Analysis." *Clinical Psychology Review* 29: 163–178.

Burt, S. Alexandra, Matt McGue, Latanya A. Carter, and William G. Iacono. 2007. "The Different Origins of Stability and Change in Antisocial Personality Disorder Symptoms." *Psychological Medicine* 37: 27–38.

Button, Tanya M. M., Jennifer Y. F. Lau, Barbara Maughan, and T. C. Eley. 2008. "Parental Punitive Discipline, Negative Life Events and Gene–Environment Interplay in the Development of Externalizing Behavior." *Psychological Medicine* 38: 29–39.

Button, Tanya Maria May, Jane Scourfield, Neilson Martin, Shaun Purcell, and Peter McGuffin. 2005. "Family Dysfunction Interacts with Genes in the Causation of Antisocial Symptoms." *Behavior Genetics* 35: 115–120.

Caspi, Avshalom, Joseph McClay, Terrie E. Moffitt, Jonathan Mill, Judy Martin, Ian W. Craig, Alan Taylor, and Richie Poulton. 2002. "Role of Genotype in the Cycle of Violence in Maltreated Children." *Science* 297:851–854.

Cauffman, Elizabeth, Laurence Steinberg, and Alex R. Piquero. 2005. "Psychological, Neuropsychological and Physiological Correlates of Serious Antisocial Behavior in Adolescence: The Role of Self-Control." *Criminology* 43: 133–176.

Chance, Susan E., Ronald T. Brown, James M. Dabbs Jr., and Robert Casey. 2000. "Testosterone, Intelligence and Behavior Disorders in Young Boys." *Personality and Individual Differences* 28: 437–445.

Page 20 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Choy, Olivia, Adrian Raine, Jill Portnoy, Anna Rudo-Hutt, Yu Gao, and Liana Soyfer. 2015. "The Mediating Role of Heart Rate on the Social Adversity-Antisocial Behavior Relationship." *Journal of Research in Crime and Delinquency* 52: 303–341.

Cima, Maaike, Tom Smeets, and Marko Jelicic. 2008. "Self-Reported Trauma, Cortisol Levels, and Aggression in Psychopathic and Non-Psychopathic Prison Inmates." *Biological Psychology* 78: 75–86.

Cohn, M. D., A. Popma, W. van den Brink, L. E. Pape, M. Kindt, L. van Domburgh, T. A. H. Doreleijers, and D. J. Veltman. 2013. "Fear Conditioning, Persistence of Disruptive Behavior and Psychopathic Traits: An fMRI Study." *Translational Psychiatry* 3: e319.

Craig, M. C., M. Catani, Q. Deeley, R. Latham, E. Daly, R. Kanaan, M. Picchioni, P. K. McGuire, T. Fahy, and D. G. M. Murphy. 2009. "Altered Connections on the Road to Psychopathy." *Molecular Psychiatry* 14: 946–953.

Cummings, E. Mark, Mona El-Sheikh, Chrystyna D. Kouros, and Peggy S. Keller. 2007. "Children's Skin Conductance Reactivity as a Mechanism of Risk in the Context of Parental Depressive Symptoms." *Journal of Child Psychology and Psychiatry* 48: 436–445.

Day, David M., Sonya G. Wanklyn, and Annie K. Yessine. 2014. "A Review of Terminological, Conceptual, and Methdological Issues in the Developmental Risk Factor Literature for Antisocial and Delinquent Behavior." *Child and Youth Care Forum* 43: 97–112.

(p. 345) De Brito, Stephane A., Essi Viding, Veena Kumari, Nigel Blackwood, and Sheilagh Hodgins. 2013. "Cool and Hot Executive Function Impairments in Violent Offenders with Antisocial Personality Disorder With and Without Psychopathy." *PloS One* 8: e65566.

Dick, Danielle M. 2011. "Gene-Environment Interaction in Psychological Traits and Disorders." *Annual Review of Clinical Psychology* 7: 383–409.

Dick, Danielle M., Richard Viken, Shaun Purcell, Jaakko Kaprio, Lea Pulkkinen, and Richard J. Rose. 2007. "Parental Monitoring Moderates the Importance of Genetic and Environmental Influences on Adolescent Smoking." *Journal of Abnormal Psychology* 116: 213–218.

Dietrich, Kim N., Ris M. Douglas, Paul A. Succop, Omer G. Berger, and Robert L. Bornschein. 2001. "Early Exposure to Lead and Juvenile Delinquency." *Neurotoxicology and Teratology* 23: 511–518.

Dolan, Margaret. 2012. "The Neuropsychology of Prefrontal Function in Antisocial Personality Disordered Offenders with Varying Degrees of Psychopathy." *Psychological Medicine* 42: 1715–1725.

Dorn, Lorah D., David J. Kolko, Elizabeth J. Susman, Bin Huang, Howard Stein, Edvin Music, and Oscar G. Bukstein. 2009. "Salivary Gonadal and Adrenal Hormone Differences in

Page 21 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Boys and Girls With and Without Disruptive Behavior Disorders: Contextual Variants." *Biological Psychology* 81: 31–39.

Edmiston, Erin E., Fei Wang, Carolyn M. Mazure, Joanne Guiney, Rajita Sinha, Linda C. Mayes, and Hilary P. Blumberg. 2011. "Corticostriatal-Limbic Gray Matter Morphology in Adolescents with Self-Reported Exposure to Childhood Maltreatment." *Archives of Pediatrics & Adolescent Medicine* 165: 1069–1077.

Ermer, Elsa, Lora M. Cope, Prashanth K. Nyalakanti, Vince D. Calhoun, and Kent A. Kiehl. 2012. "Aberrant Paralimbic Gray Matter in Criminal Psychopathy." *Journal of Abnormal Psychology* 121: 649–658.

Eysenck, Hans. J. 1977. Crime and Personality. London: Routledge & Kegan Paul.

Fagan, Shawn E., Wei Zhang, and Yu Gao. 2017. "Social Adveristy and Antisocial Behavior: Mediating Effects of Autonomic Nervous System Activity." *Journal of Abnormal Child Psychology* 45: 1–12.

Fairchild, Graeme, Cindy C. Hagan, Nicholas D. Walsh, Luca Passamonti, Andrew J. Calder, and Ian M. Goodyer. 2013. "Brain Structure Abnormalities in Adolescent Girls with Conduct Disorder." *Journal of Child Psychology and Psychiatry* 54: 86–95.

Fairchild, Graeme, Luca Passamonti, Georgina Hurford, Cindy C. Hagan, Elisabeth AH von dem Hagen, Stephanie H. M. van Goozen, Ian M. Goodyer, and Andrew J. Calder. 2011. "Brain Structure Abnormalities in Early-Onset and Adolescent-Onset Conduct Disorder." *American Journal of Psychiatry* 168: 624–633.

Fairchild, Graeme, Nicola Toschi, Kate Sully, Edmund J. S. Sonuga-Barke, Cindy C. Hagan, Stefano Diciotti, Ian M. Goodyer, Andrew J. Calder, and Luca Passamonti. 2016. "Mapping the Structural Organization of the Brain in Conduct Disorder: Replication of Findings in Two Independent Samples." *Journal of Child Psychology and Psychiatry* 57: 1018–1026.

Fairchild, Graeme, Stephanie H. M. van Goozen, Sarah J. Stollery, Jamie Brown, Julian Gardiner, Joe Herbert, and Ian M. Goodyer. 2008. "Cortisol Diurnal Rhythm and Stress Reactivity in Male Adolescents with Early-Onset or Adolescence-Onset Conduct Disorder." *Biological Psychiatry* 64: 599–606.

Farrington, David P. 1997. "The Relationship Between Low Resting Heart Rate and Violence." *Biosocial Bases of Violence*, edited by Adrian Raine, Patricia A. Brennan, David P. Farrington, and Sarnoff A. Mednick, 89–105. New York: Plenum.

(p. 346) Farrington, David P. 2005a. "The Integrated Cognitive Antisocial Potential (ICAP) Theory." In *Integrated Developmental & Life-Course Theories of Offending: Advances in Criminological Theory*, Vol. 14, edited by David P. Farrington, 73–92. New Brunswick, NJ: Transaction.

Page 22 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Farrington, David P. 2005b. "Introduction to the Integrated Developmental and Life-Course Theories of Offending." In *Integrated Developmental & Life-Course Theories of Offending: Advances in Criminological Theory*, Vol. 14, edited by David P. Farrington, 1–14. New Brunswick, NJ: Transaction.

Fast, Diane K., Julianne Conry, and Christine A. Loock. 1999. "Identifying Fetal Alcohol Syndrome Among Youth in the Criminal Justice System." *Journal of Developmental & Behavioral Pediatrics* 20: 370–372.

Firestone, Philip, and Susan Peters. 1983. "Minor Physical Anomalies and Behavior in Children: A Review." *Journal of Autism and Developmental Disorders* 13: 411–425.

Freedman, Lauren F., and Simon N. Verdun-Jones. 2010. "Blaming the Parts Instead of the Person: Understanding and Applying Neurobiological Factors Associated with Psychopathy." *Canadian Journal of Criminology and Criminal Justice* 52: 29–53.

Gao, Yu, Laura A. Baker, Adrian Raine, Henry Wu, and Serena Bezdjian. 2009. "Brief Report: Interaction Between Social Class and Risky Decision-Making in Children with Psychopathic Tendencies." *Journal of Adolescence* 32: 409–414.

Gao, Yu, Adrian Raine, Peter H. Venables, Michael E. Dawson, and Sarnoff A. Mednick. 2010. "Association of Poor Childhood Fear Conditioning and Adult Crime." *The American Journal of Psychiatry* 167: 56–60.

Gao, Yu, Catherine Tuvblad, Anne Schell, Laura Baker, and Adrian Raine. 2015. "Skin Conductance Fear Conditioning Impairments and Aggression: A Longitudinal Study." *Psychophysiology* 52: 288–295.

Gibson, Chris L., and Stephen G. Tibbetts. 2000. "A Biosocial Interaction in Predicting Early Onset of Offending." *Psychological Reports* 86: 509–518.

Glenn, Andrea L., and Adrian Raine. 2014. "Neurocriminology: Implications for the Punishment, Prediction and Prevention of Criminal Behaviour." *Nature Reviews Neuroscience* 15: 54–63.

Glenn, Andrea L., Adrian Raine, Robert A. Schug, Yu Gao, and Douglas A. Granger. 2011. "Increased Testosterone-to-Cortisol Ratio in Psychopathy." *Journal of Abnormal Psychology* 120: 389–999.

Gray, Peter B., Sonya M. Kahlenberg, Emily S. Barrett, Susan F. Lipson, and Peter T. Ellison. 2002. "Marriage and Fatherhood Are Associated with Lower Testosterone in Males." *Evolution and Human Behavior* 23: 193–201.

Guo, Guang, Michael E. Roettger, and Jean C. Shih. 2007. "Contributions of the DAT1 and DRD2 Genes to Serious and Violent Delinquency Among Adolescents and Young Adults." *Human Genetics* 121: 125–136.

Page 23 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Haberstick, Brett C., Stephanie Schmitz, Susan E. Young, and John K. Hewitt. 2006. "Genes and Developmental Stability of Aggressive Behavior Problems at Home and School in a Community Sample of Twins Aged 7–12." *Behavior Genetics* 36: 809–819.

Hancock, Megan, Jennifer L. Tapscott, and Peter NS Hoaken. 2010. "Role of Executive Dysfunction in Predicting Frequency and Severity of Violence." *Aggressive Behavior* 36: 338–349.

Hanlon, Robert E., Joseph J. Coda, Derin Cobia, and Leah H. Rubin. 2012. "Psychotic Domestic Murder: Neuropsychological Differences Between Homicidal and Nonhomicidal Schizophrenic Men." *Journal of Family Violence* 27: 105–113.

(p. 347) Hanlon, Robert E., Leah H. Rubin, Marie Jensen, and Sarah Daoust. 2010. "Neuropsychological Features of Indigent Murder Defendants and Death Row Inmates in Relation to Homicidal Aspects of Their Crimes." *Archives of Clinical Neuropsychology* 25: 1–13.

Hanna, Gregory L., Arthur Yuwiler, and Janice K. Coates. 1995. "Whole Blood Serotonin and Disruptive Behaviors in Juvenile Obsessive-Compulsive Disorder." *Journal of the American Academy of Child & Adolescent Psychiatry* 34: 28–35.

Hawes, David J., John Brennan, and Mark R. Dadds. 2009. "Cortisol, Callous-Unemotional Traits, and Pathways to Antisocial Behavior." *Current Opinion in Psychiatry* 22: 357–362.

Hicks, Brian M., Marie D. Carlson, Daniel M. Blonigen, Christopher J. Patrick, William G. Iacono, and Matt MGue. 2012. "Psychopathic Personality Traits and Environmental Contexts: Differential Correlates, Gender Differences, and Genetic Mediation." *Personality Disorders: Theory, Research, and Treatment* 3: 209–227.

Hirata, Yuko, Clement C. Zai, Behdin Nowrouzi, Joseph H. Beitchman, and James L. Kennedy. 2013. "Study of the Catechol-O-Methyltransferase (COMT) Gene with High Aggression in Children." *Aggressive Behavior* 39: 45–51.

Hodgins, Sheilagh, Lynn Kratzer, and Thomas F. McNeil. 2001. "Obstetric Complications, Parenting, and Risk of Criminal Behavior." *Archives of General Psychiatry* 58(8): 746–752.

Holi, Matti, Laura Auvinen-Lintunen, Nina Lindberg, Pekka Tani, and Matti Virkkunen. 2006. "Inverse Correlation Between Severity of Psychopathic Traits and Serum Cortisol Levels in Young Adult Violent Male Offenders." *Psychopathology* 39: 102–104.

Hugdahl, Kenneth. 2001. Psychophysiology. Cambridge, MA: Harvard University Press.

Hughes, Carroll W., Frederick Petty, Sabri Sheikha, and Gerald L. Kramer. 1996. "Whole-Blood Serotonin in Children and Adolescents with Mood and Behavior Disorders." *Psychiatry Research* 65: 79–95.

Isen, Joshua. 2010. "A Meta-Analytic Assessment of Wechsler's P>V Sign in Antisocial Populations." *Clinical Psychology Review* 30: 423–435.

Page 24 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Ishikawa, Sharon S., Adrian Raine, Todd Lencz, Susan Bihrle, and Lori Lacasse. 2001. "Autonomic Stress Reactivity and Executive Functions in Successful and Unsuccessful Criminal Psychopaths from the Community." *Journal of Abnormal Psychology* 110: 423– 432.

Jennings, Wesley G., Alex R. Piquero, and David P. Farrington. 2013. "Does Resting Heart Rate at Age 18 Distinguish General and Violent Offending up to Age 50? Findings from the Cambridge Study in Delinquent Development." *Journal of Criminal Justice* 41: 213– 219.

Kandel, Elizabeth, Patricia A. Brennan, Sarnoff A. Mednick, and N. M. Michelson. 1989. "Minor Physical Anomalies and Recidivistic Adult Violent Criminal Behavior." *Acta Psychiatrica Scandinavica* 79(1): 103–107.

Kandel, Elizabeth, and Sarnoff A. Mednick. 1991. "Perinatal Complications Predict Violent Offending." *Criminology* 29: 519–529.

Kemppainen, Liisa, Jari Jokelainen, Marjo-Riitta Järvelin, Matti Isohanni, and Pirkko Räsänen. 2001. "The One-Child Family and Violent Criminality: A 31-Year Follow-Up Study of the Northern Finland 1966 Birth Cohort." *American Journal of Psychiatry* 158: 960–962.

Koopmans, Judith R., Wendy S. Slutske, G. Caroline M. Van Baal, and Dorret I. Boomsma. 1999. "The Influence of Religion on Alcohol Use Initiation: Evidence for Genotype X Environment Interaction." *Behavior Genetics* 29: 445–453.

Kosson, David S., Sarah K. Miller, Katherine A. Byrnes, and Catherine L. Leveroni. 2007. "Testing Neuropsychological Hypotheses for Cognitive Deficits in Psychopathic Criminals: A Study of Global-Local Processing." *Journal of the International Neuropsychological Society* 13: 267–276.

(p. 348) Kruesi, Markus J. P., Susan Swedo, Henrietta Leonard, David R. Rubinow, and Judith L. Rapoport. 1990. "CSF Somatostatin in Childhood Psychiatric Disorders: A Preliminary Investigation." *Psychiatry Research* 33: 277–284.

Lahey, Benjamin B., and Irwin D. Waldman. 2005. "A Developmental Model of the Propensity to Offend During Childhood and Adolescence." In *Integrated Developmental & Life-Course Theories of Offending: Advances in Criminological Theory,* Vol. 14, edited by David P. Farrington, 15–50. New Brunswick, NJ: Transaction.

Latvala, Antti, Ralf Kuja-Halkola, Catarina Almqvist, Henrik Larsson, and Paul Lichtenstein. 2015. "Heart Rate and Violent Criminality in More Than 700,000 Men." *JAMA Psychiatry* 72: 971–978.

Liu, Huasheng, Jian Liao, Weixiong Jiang, and Wei Wang. 2014. "Changes in Low-Frequency Fluctuations in Patients with Antisocial Personality Disorder Revealed by Resting-State Functional MRI." *PloS One* 9: e89790.

Page 25 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Liu, Jianghong, Adrian Raine, Anne Wuerker, Peter H. Venables, and Sarnoff Mednick. 2009. "The Association of Birth Complications and Externalizing Behavior in Early Adolescents: Direct and Mediating Effects." *Journal of Research on Adolescence* 19:93–111.

Loney, Bryan R., Paul J. Frick, Mesha Ellis, and Monique G. McCoy. 1998. "Intelligence, Callous-Unemotional Traits, and Antisocial Behavior." *Journal of Psychopathology and Behavioral Assessment* 20: 231–247.

Lorber, Michael F. 2004. "Psychophysiology of Aggression, Psychopathy, and Conduct Problems: A Meta-Analysis." *Psychological Bulletin* 130: 531–552.

Ly, Martina, Julian C. Motzkin, Carissa L. Philippi, Gregory R. Kirk, Joseph P. Newman, Kent A. Kiehl, and Michael Koenigs. 2012. "Cortical Thinning in Psychopathy." *American Journal of Psychiatry* 169: 743-749.

Malik, Ayesha I., Clement C. Zai, Zihad Abu, Behdin Nowrouzi, and Joseph H. Beitchman. 2012. "The Role of Oxytocin and Oxytocin Receptor Gene Variants in Childhood-Onset Aggression." *Genes, Brain and Behavior* 11: 545–551.

Maras, Athanasios, Manfred Laucht, Dirk Gerdes, Cindy Wilhelm, Sabina Lewicka, Doris Haack, Lucie Malisova, and Martin H. Schmidt. 2003. "Association of Testosterone and Dihydrotestosterone with Externalizing Behavior in Adolescent Boys and Girls." *Psychoneuroendocrinology* 28: 932–940.

Mattson, Sarah N., and Edward P. Riley. 2000. "Parent Ratings of Behavior in Children with Heavy Prenatal Alcohol Exposure and IQ-Matched Controls." *Alcoholism: Clinical and Experimental Research* 24: 226–231.

Mazur, Allan. 1995. "Biosocial Models of Deviant Behavior Among Male Army Veterans." *Biological Psychology* 41: 271–293.

McBurnett, Keith, Benjamin B. Lahey, Paul J. Rathouz, and Rolf Loeber. 2000. "Low Salivary Cortisol and Persistent Aggression in Boys Referred for Disruptive Behavior." *Archives of General Psychiatry* 57: 38–43.

Mednick, Sarnoff A., and Elizabeth S. Kandel. 1988. "Congenital Determinants of Violence." *Journal of the American Academy of Psychiatry and the Law Online* 16: 101–109.

Middeldorp, Christel M., Diane J. Lamb, Jacqueline M. Vink, Meike Bartels, Catharina E. M. van Beijsterveldt, and Dorret I. Boomsma. 2014. "Child Care, Socio-Economic Status and Problem Behavior: A Study of Gene-Environment Interaction in Young Dutch Twins." *Behavior Genetics* 44: 314–325.

Moffitt, Terrie E. 1993. "Adolescence-Limited and Life-Course-Persistent Antisocial Behavior: A Developmental Taxonomy." *Psychological Review* 100: 674–701.

(p. 349) Moffitt, Terrie E., Donald R. Lynam, and Phil A. Silva. 1994. "Neuropsychological Tests Predicting Persistent Male Delinquency." *Criminology* 32: 277–300.

Page 26 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Moore, Todd M., Angela Scarpa, and Adrian Raine. 2002. "A Meta-Analysis of Serotonin Metabolite 5-HIAA and Antisocial Behavior." *Aggressive Behavior* 28: 299–316.

Morgan, Alex B., and Scott O. Lilienfeld. 2000. "A Meta-Analytic Review of the Relation Between Antisocial Behavior and Neuropsychological Measures of Executive Function." *Clinical Psychology Review* 20: 113–136.

Moul, Caroline, Carol Dobson-Stone, John Brennan, David Hawes, and Mark Dadds. 2013. "An Exploration of the Serotonin System in Antisocial Boys with High Levels of Callous-Unemotional Traits." *PloS One* 8: e56619.

Nestor, Paul G. 1992. "Neuropsychological and Clinical Correlates of Murder and Other Forms of Extreme Violence in a Forensic Psychiatric Population." *The Journal of Nervous and Mental Disease* 180: 418–423.

Nigg, Joel T., Jennifer M. Glass, Maria M. Wong, Edwin Poon, Jennifer M. Jester, Hiram E. Fitzgerald, Leon I. Puttler, Kenneth M. Adams, and Robert A. Zucker. 2004. "Neuropsychological Executive Functioning in Children at Elevated Risk for Alcoholism: Findings in Early Adolescence." *Journal of Abnormal Psychology* 113: 302–314.

O'Leary, Megan M., Bryan R. Loney, and Lisa A. Eckel. 2007. "Gender Differences in the Association Between Psychopathic Personality Traits and Cortisol Response to Induced Stress." *Psychoneuroendocrinology* 32: 183–191.

Olson, Heather Carmichael, Ann P. Streissguth, Paul D. Sampson, Helen M. Barr, Fred L. Bookstein, and Keith Thiede. 1997. "Association of Prenatal Alcohol Exposure with Behavioral and Learning Problems in Early Adolescence." *Journal of the American Academy of Child & Adolescent Psychiatry* 36: 1187–1194.

Oostermeijer, Sanne, Sarah Whittle, Chao Suo, Nicholas B. Allen, Julian G. Simmons, Nandita Vijayakumar, Peter M. van de Ven, Lucres M. C. Jansen, Murat Yücel, and Arne Popma. 2016. "Trajectories of Adolescent Conduct Problems in Relation to Cortical Thickness Development: A Longitudinal MRI Study." *Translational Psychiatry* 6: e841.

Ortiz, Jame, and Adrian Raine. 2004. "Heart Rate Level and Antisocial Behavior in Children and Adolescents: A Meta-Analysis." *Journal of the American Academy of Child & Adolescent Psychiatry* 43: 154–162.

Paradis, Angela D., Garrett M. Fitzmaurice, Karestan C. Koenen, and Stephen L. Buca. 2015. "A Prospective Investigation of Neurodevelopmental Risk Factors for Adult Antisocial Behavior Combining Official Arrest Records and Self-Reports." *Journal of Psychiatric Research* 68: 363–370.

Pardini, Dustin A., Adrian Raine, Kirk Erickson, and Rolf Loeber. 2014. "Lower Amygdala Volume in Men Is Associated with Childhood Aggression, Early Psychopathic Traits, and Future Violence." *Biological Psychiatry* 75: 73–80.

Page 27 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Pine, Daniel S., David Shaffer, Irvin Sam Schonfeld, and Mark Davies. 1997. "Minor Physical Anomalies: Modifiers of Environmental Risks for Psychiatric Impairment?" *Journal of the American Academy of Child & Adolescent Psychiatry* 36: 395–403.

Platje, Evelien, Lucres Jansen, Adrian Raine, Susan J. T. Branje, Theo A. H. Doreleijers, Marjan de Vries-Bouw, Arne Popma, et al. 2013. "Longitudinal Associations in Adolescence Between Cortisol and Persistent Aggressive or Rule-Breaking Behavior." *Biological Psychology* 93: 132–137.

Pope, Harrison G., Elena M. Kouri, and James I. Hudson. 2000. "Effects of Supraphysiologic Doses of Testosterone on Mood and Aggression in Normal Men: A Randomized Controlled Trial." *Archives of General Psychiatry* 57: 133–140.

(p. 350) Portnoy, Jill, Adrian Raine, Frances R. Chen, Dustin Pardini, Rolf Loeber, and J. Richard Jennings. 2014. "Heart Rate and Antisocial Behavior: The Mediating Role of Impulsive Sensation Seeking." *Criminology* 52: 292–311.

Portnoy, Jill and David P. Farrington. 2015. "Resting Heart Rate and Antisocial Behavior: An Updated Systematic Review and Meta-analysis." *Aggression and Violent Behavior* 22: 33–45.

Quay, Herbert C. 1965. "Psychopathic Personality as Pathological Stimulation-Seeking." *American Journal of Psychiatry* 122: 180–183.

Raine, Adrian. 1993. *The Psychopathology of Crime: Criminal Behavior as a Clinical Disorder*. San Diego, CA: Academic Press.

Raine, Adrian. 2002a. "Annotation: The Role of Prefrontal Deficits, Low Autonomic Arousal, and Early Health Factors in the Development of Antisocial and Aggressive Behavior in Children." *Journal of Child Psychology and Psychiatry* 43: 417–434.

Raine, Adrian. 2002b. "Biosocial Studies of Antisocial and Violent Behavior in Children and Adults: A Review." *Journal of Abnormal Child Psychology* 30: 311–326.

Raine, Adrian, Annis Lai Chu Fung, Jill Portnoy, Olivia Choy, and Victoria L. Spring. 2014. "Low Heart Rate as a Risk Factor for Child and Adolescent Proactive Aggressive and Impulsive Psychopathic Behavior." *Aggressive Behavior* 40: 290–299.

Raine, Adrian, Patricia Brennan, and Sarnoff A. Mednick. 1994. "Birth Complications Combined with Early Maternal Rejection at Age 1 Year Predispose to Violent Crime at Age 18 Years." *Archives of General Psychiatry* 51: 984–988.

Raine, Adrian, Patricia Brennan, and Sarnoff A. Mednick. 1997. "Interaction Between Birth Complications and Early Maternal Rejection in Predisposing Individuals to Adult Violence: Specificity to Serious, Early-Onset Violence." *American Journal of Psychiatry* 154: 1265–1271.

Page 28 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Raine, Adrian, William S. Laufer, Yaling Yang, Katherine L. Narr, Paul Thompson, and Arthur W. Toga. 2012. "Increased Executive Functioning, Attention, and Cortical Thickness in White-Collar Criminals." *Human Brain Mapping* 33: 2932–2940.

Raine, Adrian, Todd Lencz, Susan Bihrle, Lori LaCasse, and Patrick Colletti. 2000. "Reduced Prefrontal Gray Matter Volume and Reduced Autonomic Activity in Antisocial Personality Disorder." *Archives of General Psychiatry* 57: 119–127.

Raine, Adrian, Todd Lencz, Kristen Taylor, Joseph B. Hellige, Susan Bihrle, Lori Lacasse, Mimi Lee, Sharon Ishikawa, and Patrick Colletti. 2003. "Corpus Callosum Abnormalities in Psychopathic Antisocial Individuals." *Archives of General Psychiatry* 60: 1134–1142.

Raine, Adrian, Terrie E. Moffitt, Avshalom Caspi, Rolf Loeber, Magda Stouthamer-Loeber, and Don Lynam. 2005. "Neurocognitive Impairments in Boys on the Life-Course Persistent Antisocial Path." *Journal of Abnormal Psychology* 114: 38–49.

Raine, Adrian, Sohee Park, Todd Lencz, Susan Bihrle, Lori LaCasse, Cathy Spatz Widom, Louai Al-Dayeh, and Manbir Singh. 2001. "Reduced Right Hemisphere Activation in Severely Abused Violent Offenders During a Working Memory Task: An fMRI Study." *Aggressive Behavior* 27: 111–129.

Raine, Adrian, Jacqueline Stoddard, Susan Bihrle, and Monte Buchsbaum. 1998. "Prefrontal Glucose Deficits in Murderers Lacking Psychosocial Deprivation." *Cognitive and Behavioral Neurology* 11: 1–7.

Raine, Adrian, Peter H. Venables, and Sarnoff A. Mednick. 1997. "Low Resting Heart Rate at Age 3 Years Predisposes to Aggression at Age 11 Years: Evidence from the Mauritius Child Health Project." *Journal of the American Academy of Child & Adolescent Psychiatry* 36: 1457–1464.

(p. 351) Raine, Adrian, Peter H. Venables, and Mark Williams. 1990. "Relationships Between Central and Autonomic Measures of Arousal at Age 15 Years and Criminality at Age 24 Years." *Archives of General Psychiatry* 47: 1003–1007.

Raine, Adrian, Peter H. Venables, and Mark Williams. 1995. "High Autonomic Arousal and Electrodermal Orienting at Age 15 Years as Protective Factors Against Criminal Behavior at Age 29 Years." *American Journal of Psychiatry* 152: 1595–1600.

Raine, Adrian, Peter H. Venables, and Mark Williams. 1996. "Better Autonomic Conditioning and Faster Electrodermal Half-Recovery Time at Age 15 Years as Possible Protective Factors Against Crime at Age 29 Years." *Developmental Psychology* 32: 624–630.

Raine, Adrian, Pauline S. Yaralian, Chandra Reynolds, Peter H. Venables, and Sarnoff A. Mednick. 2002c. "Spatial but Not Verbal Cognitive Deficits at Age 3 Years in Persistently Antisocial Individuals." *Development and Psychopathology* 14: 25–44.

Page 29 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Räsänen, Pirkko, Helinä Hakko, Matti Isohanni, Sheilagh Hodgins, Marjo-Riitta Järvelin, and Jari Tiihonen. 1999. "Maternal Smoking During Pregnancy and Risk of Criminal Behavior Among Adult Male Offspring in the Northern Finland 1966 Birth Cohort." *American Journal of Psychiatry* 156: 857–862.

Rasmussen, Kirsten, Roger Almvik, and Sten Levander. 2001. "Performance and Strategy Indices of Neuropsychological Tests: Relations with Personality, Criminality and Violence." *Journal of Forensic Neuropsychology* 2: 29–43.

Rhee, Soo Hyun, and Irwin D. Waldman. 2002. "Genetic and Environmental Influences on Antisocial Behavior: A Meta-Analysis of Twin and Adoption Studies." *Psychological Bulletin* 128: 490–529.

Rogers, Jack C., and Stéphane A. De Brito. 2016. "Cortical and Subcortical Gray Matte Volume in Youths with Conduct Problems: A Meta-Analysis." *JAMA Psychiatry* 73: 64–72.

Rolls, Edmund T. 2000. "The Orbitofrontal Cortex and Reward." *Cerebral Cortex* 10: 284–294.

Rose, Richard J., Danielle M. Dick, Richard J. Viken, and Jaakko Kaprio. 2001. "Gene-Environment Interaction in Patterns of Adolescent Drinking: Regional Residency Moderates Longitudinal Influences on Alcohol Use." *Alcoholism: Clinical and Experimental Research* 25: 637–643.

Rowe, David C., David M. Almeida, and Kristen C. Jacobson. 1999. "School Context and Genetic Influences on Aggression in Adolescence." *Psychological Science* 10: 277–280.

Salekin, Randall T., Craig S. Neumann, Anne-Marie R. Leistico, and Alecia A. Zalot. 2004. "Psychopathy in Youth and Intelligence: An Investigation of Cleckley's Hypothesis." *Journal of Clinical Child and Adolescent Psychology* 33: 731–742.

Schiffer, Boris, Bernhard W. Müller, Norbert Scherbaum, Sheilagh Hodgins, Michael Forsting, Jens Wiltfang, Elke R. Gizewski, and Norbert Leygraf. 2011. "Disentangling Structural Brain Alterations Associated with Violent Behavior from Those Associated with Substance Use Disorders." *Archives of General Psychiatry* 68: 1039–1049.

Schiffer, Boris, Christina Pawliczek, Michael Forsting, Elke Gizewski, Norbert Leygraf, Sheilagh Hodgins, et al. 2014. "Neural Mechanisms Underlying Cognitive Control of Men with Lifelong Antisocial Behavior." *Psychiatry Research: Neuroimaging* 222: 43–51.

Schiffer, Boris, and Corinne Vonlaufen. 2011. "Executive Dysfunctions in Pedophilic and Nonpedophilic Child Molesters." *The Journal of Sexual Medicine* 8: 1975–1984.

Sijtsema, Jelle J., René Veenstra, Siegwart Lindenberg, Arie M. van Roon, Frank C. Verhulst, Johan Ormel, and Harriëtte Riese. 2010. "Mediation of Sensation Seeking and Behavioral (p. 352) Inhibition on the Relationship Between Heart Rate and Antisocial Behav-

Page 30 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

ior: The TRAILS Study." Journal of the American Academy of Child & Adolescent Psychiatry 49: 493–502.

Soderstrom, Hendrik, Kaj Blennow, Anna Manhem, and A. Forsman. 2001. "CSF Studies in Violent Offenders." *Journal of Neural Transmission* 108: 869–878.

Soderstrom, Hendrik, Kaj Blennow, A. K. Sjodin, and A. Forsman. 2003. "New Evidence for an Association Between the CSF HVA: 5-HIAA Ratio and Psychopathic Traits." *Journal of Neurology, Neurosurgery & Psychiatry* 74: 918–921.

Sondeijker, Frouke E. P. L., Robert F. Ferdinand, Albertine J. Oldehinkel, Henning Tiemeier, Johan Ormel, and Frank C. Verhulst. 2008. "HPA-Axis Activity as a Predictor of Future Disruptive Behaviors in Young Adolescents." *Psychophysiology* 45: 398–404.

Stålenheim, E. Gunilla, Elias Eriksson, Lars von Knorring, and Leif Wide. 1998. "Testosterone as a Biological Marker in Psychopathy and Alcoholism." *Psychiatry Research* 77: 79–88.

Stanford, Matthew S., Sarah M. Conklin, Laura E. Helfritz, and Tim R. Kockler. 2007. "P3 Amplitude Reduction and Executive Function Deficits in Men Convicted of Spousal/Partner Abuse." *Personality and Individual Differences* 43: 365–375.

Tarter, Ralph E., Levent Kirisci, Judith S. Gavaler, Maureen Reynolds, Galina Kirillova, Duncan B. Clark, Jionglin Wu, Howard B. Moss, and Michael Vanyukov. 2009. "Prospective Study of the Association Between Abandoned Dwellings and Testosterone Level on the Development of Behaviors Leading to Cannabis Use Disorder in Boys." *Biological Psychiatry* 65: 116–121.

Taylor, Jeanette, William G. Iacono, and Matt McGue. 2000. "Evidence for a Genetic Etiology of Early-Onset Delinquency." *Journal of Abnormal Psychology* 109: 634–643.

Tellegen, Auke, and Niels G. Waller. 2008. "Exploring Personality Through Test Construction: Development of the Multidimensional Personality Questionnaire." In *The SAGE Handbook of Personality Theory and Assessment*, Vol. 2, edited by Gregory J. Boyle, Gerald Matthews, and Donald H. Saklofske, 261–292. London: Sage.

Tuvblad, Catherine. 2013. "Genetic Influences on Antisocial Behavior over the Life Course." In *Routledge Handbook of Biosocial Criminology*, edited by Matt DeLisi and Michael G. Vaughn, 77-100. New York: Routledge.

Tuvblad, Catherine, Martin Grann, and Paul Lichtenstein. 2006. "Heritability for Adolescent Antisocial Behavior Differs with Socioeconomic Status: Gene–Environment Interaction." *Journal of Child Psychology and Psychiatry* 47: 734–743.

Tuvblad, Catherine, Jurgita Narusyte, Martin Grann, Jerzy Sarnecki, and Paul Lichtenstein. 2011. "The Genetic and Environmental Etiology of Antisocial Behavior from Childhood to Emerging Adulthood." *Behavior Genetics* 41: 629–640.

Page 31 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

van Goozen, Stephanie H. M., Graeme Fairchild, Heddeke Snoek, and Gordon T. Harold. 2007. "The Evidence for a Neurobiological Model of Childhood Antisocial Behavior." *Psychological Bulletin* 133: 149–182.

Van Goozen, Stephanie H. M., Walter Matthys, Peggy T. Cohen-Kettenis, Jos H. H. Thijssen, and Herman van Engeland. 1998. "Adrenal Androgens and Aggression in Conduct Disorder Prepubertal Boys and Normal Controls." *Biological Psychiatry* 43: 156–158.

Van Bokhoven, Irene, Stephanie H. M. Van Goozen, Herman Van Engeland, Benoist Schaal, Louise Arseneault, Jean R. Séguin, Jean-Marc Assaad, Daniel S. Nagin, Frank Vitaro, and Richard E. Tremblay. 2006. "Salivary Testosterone and Aggression, Delinquency, and Social Dominance in a Population-Based Longitudinal Study of Adolescent Males." *Hormones and Behavior* 50: 118–125.

(p. 353) Van Honk, Jack, Dennis J. Schutter, Peter A. Bos, Anne-Wil Kruijt, Eef G. Lentjes, and Simon Baron-Cohen. 2011. "Testosterone Administration Impairs Cognitive Empathy in Women Depending on Second-to-Fourth Digit Ratio." *Proceedings of the National Academy of Sciences* 108: 3448–3452.

Vaske, Jamie, Jamie Newsome, and Danielle Boisvert. 2013. "The Mediating Effects of Verbal Skills in the Relationship Between Low Birth Weight and Childhood Aggressive Behaviour." *Infant and Child Development* 22: 235–249.

Vaske, Jamie, Jamie Newsome, Danielle L. Boisvert, Alex R. Piquero, Angela D. Paradis, and Stephen L. Buka. 2015. "The Impact of Low Birth Weight and Maternal Age on Adult-hood Offending." *Journal of Criminal Justice* 43: 49–56.

Viding, Essi, R. James R. Blair, Terrie E. Moffitt, and Robert Plomin. 2005. "Evidence for Substantial Genetic Risk for Psychopathy in 7-Year-Olds." *Journal of Child Psychology and Psychiatry* 46: 592–597.

Viding, Essi, Paul J. Frick, and Robert Plomin. 2007. "Aetiology of the Relationship Between Callous–Unemotional Traits and Conduct Problems in Childhood." *The British Journal of Psychiatry* 190: s33–38.

Volavka, J. A. N., Robert Bilder, and Karen Nolan. 2004. "Catecholamines and Aggression: The Role of COMT and MAO Polymorphisms." *Annals of the New York Academy of Sciences* 1036: 393–398.

Wakschlag, Lauren S., Benjamin B. Lahey, Rolf Loeber, Stephanie M. Green, Rachel A. Gordon, and Bennett L. Leventhal. 1997. "Maternal Smoking During Pregnancy and the Risk of Conduct Disorder in Boys." *Archives of General Psychiatry* 54: 670–676.

Wakschlag, Lauren S., Kate E. Pickett, Edwin Cook Jr, Neal L. Benowitz, and Bennett L. Leventhal. 2002. "Maternal Smoking During Pregnancy and Severe Antisocial Behavior in Offspring: A Review." *American Journal of Public Health* 92: 966–974.

Page 32 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Wikström, Per-Olof H. (2005). "The Social Origins of Pathways in Crime: Towards a Developmental Ecological Action Theory of Crime Involvement and Its Changes. In *Integrated Developmental & Life-Course Theories of Offending: Advances in Criminological Theory,* Vol. 14, edited by David P. Farrington, 211–245. New Brunswick, NJ: Transaction.

Wilson, James Q. and Richard Herrnstein. 1985. *Crime and Human Nature*. New York: Simon & Schuster.

Wright, John Paul, Kim N. Dietrich, M. Douglas Ris, Richard W. Hornung, Stephanie D. Wessel, Bruce P. Lanphear, Mona Ho, and Mary N. Rae. 2008. "Association of Prenatal and Childhood Blood Lead Concentrations with Criminal Arrests in Early Adulthood." *PLoS Medicine* 5: e101.

Yang, Yaling, and Adrian Raine. 2009. "Prefrontal Structural and Functional Brain Imaging Findings in Antisocial, Violent, and Psychopathic Individuals: A Meta-Analysis." *Psychiatry Research: Neuroimaging* 174: 81–88.

Yang, Yaling, Adrian Raine, Patrick Colletti, Arthur W. Toga, and Katherine L. Narr. 2010a. "Morphological Alterations in the Prefrontal Cortex and the Amygdala in Unsuccessful Psychopaths." *Journal of Abnormal Psychology* 119: 546–554.

Yang, Yaling, Adrian Raine, Chen-Bo Han, Robert A. Schug, Arthur W. Toga, and Katherine L. Narr. 2010b. "Reduced Hippocampal and Parahippocampal Volumes in Murderers with Schizophrenia." *Psychiatry Research: Neuroimaging* 182: 9–13.

Yang, Yaling, Adrian Raine, Todd Lencz, Susan Bihrle, Lori LaCasse, and Patrick Colletti. 2005. "Volume Reduction in Prefrontal Gray Matter in Unsuccessful Criminal Psychopaths." *Biological Psychiatry* 57: 1103–1108.

(p. 354) Yang, Yaling, Adrian Raine, Katherine L. Narr, Patrick Colletti, and Arthur W. Toga. 2009. "Localization of Deformations Within the Amygdala in Individuals with Psychopathy." *Archives of General Psychiatry* 66: 986–994.

Yang, Yaling, Pan Wang, Laura A. Baker, Katherine L. Narr, Shantanu, H. Joshi, George Hafzalla, Adrian Raine, Paul M. Thompson. 2015. "Thicker Temporal Cortex Associates with a Developmental Trajectory for Psychopathic Traits in Adolescents." *PloS One* 10: e0127025.

Yildirim, Baris O. and Jan J. L. Derksen. 2012. "A Review on the Relationship Between Testosterone and Life-Course Persistent Antisocial Behavior." *Psychiatry Research* 200: 984–1010.

Yu, Rongqin, Jaap Nieuwenhuis, Wim Meeus, Pieter Hooimeijer, Hans M. Koot, and Susan Branje. 2016. "Biological Sensitivity to Context: Cortisol Awakening Response Moderates the Effects of Neighbourhood Density on the Development of Adolescent Externalizing Problem Behaviours." *Biological Psychology* 120: 96–107.

Page 33 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).

Zai, Clement C., Katherine E. Muir, Behdin Nowrouzi, Sajid A. Shaikh, Esther Choi, Laura Berall, Marc-Olivier Trépanier, Joseph H. Beitchman, and James L. Kennedy. 2012. "Possible Genetic Association Between Vasopressin Receptor 1B and Child Aggression." *Psychiatry Research* 200: 784–788.

Olivia Choy

Olivia Choy is an Assistant Professor of Psychology at Nanyang Technological University, Singapore.

Jill Portnoy

Jill Portnoy is an Assistant Professor in the School of Criminology and Justice Studies at the University of Massachusetts, Lowell.

Adrian Raine

Adrian Raine is the Richard Perry University Professor in the Departments of Criminology, Psychiatry, and Psychology at the University of Pennsylvania.

Rheanna J. Remmel

Rheanna J. Remmel is a graduate student in the Clinical Psychology and Law program at the University of Alabama.

Robert Schug

Robert Schug is an Associate Professor of Criminology, Criminal Justice, and Forensic Psychology at California State University, Long Beach.

Catherine Tuvblad

Catherine Tuvblad is an Assistant Professor of Criminology at Örebro University, Sweden.

Yaling Yang

Yaling Yang is an Assistant Professor of Pediatrics at Children's Hospital Los Angeles, University of Southern California.

Page 34 of 34

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). © Oxford University Press, 2018. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use (for details see Privacy Policy and Legal Notice).